Author's response to reviews

Title: SSεCKS/Gravin/AKAP12 attenuates expression of proliferative and angiogenic genes during suppression of v-Src-induced oncogenesis

Authors:

Yongzhong Liu (yoliu@mail.nih.gov)
Lingqiu Gao (lingqiu.gao@roswellpark.org)
Irwin H Gelman (irwin.gelman@roswellpark.org)

Version: 6 Date: 23 April 2006

Author's response to reviews:

Changes to the formatting were as per requested in the email of 4/21/06 from Iratxe Puebla. Specifically, the authors' affiliations were reformatted as requested, section headings were de-underlined and converted into sentence case, and all the uses of "et al." in the reference section were changed to include all authors.